## ACCELERATORS & EXPERIMENTAL FACILITIES STATUS SUMMARY OF WEEK 33 - 2023

Technical infrastructure – J. Nielsen Linac 4 – A. Lombardi PS Booster – F. Asvesta ISOLDE – E. Fadakis PS – A. Huschquer PS – East Area – N. Charitonidis PS – nTOF – M. Bacak AD – ELENA – B. Dupuy SPS – K. Li SPS – North Area – N. Charitonidis SPS – AWAKE – G. Zevi Della Porta SPS – HiRadMat – Not running, no report Linac 3 – G. Bellodi LEIR – No report LHC – J. Wenninger & M. Solfaroli CLEAR – V. Rieker

	Techr	<b>nical I</b> r	nfrastr	ucture (1	<b>FI)</b>				
Facility Coord	inator last week	Jesper Nielsen							
Facility Coord	inator this week	Jesper Nielsen							
			Statistics	5					
Alarms									
Phone calls	Incoming Outgoing								
<b>ODM</b> s									
		Fa	cility Stat	tus					
Summary	Quite a busy week	, in partic	cular with	several tech	nical issues	around LHC5!			
Issues	Tue 15/08/23 05 similar to recent p Wed 16/08/23 2 cryogenics Breake Breaker). EN-EL co required with par Wed 16/08/23 2 External. This is a magnet to quenc controls, and it pu ventilation went b system. Fri 18/08/23 05: in SH5 in LHC5. A same place. TI gu cause of the smol UPS. TI asks FireB do so in case the the UPS on bypas replaced, and UPS Very good coordi to a successful int fact that we had f Sun 20/08/23 15 building 279. A pl motor. Piquet was caused a stop also	<b>i.i.</b> particular (i.i. particular (i.i. particular (i.i. particular) (i.i. parti	connecti connecti Best effo S Coldbo on EKD200 reset or s down of ctrical ala bard conti erlock wa ntilation i ominal af ce alarms ne time al Brigade to d the Fire o not cut f evolves, cut CRYO ed withou etween Fi n with no tion arou ling towe a fault of ed for int O installa	on problem ort called an ox disconnec 8/5E alarm r witch back r CMS to low rm on EXD1 colled by CM s sent via DS n flush mod ter a reset b seen on the arms seen c owards the U Brigade cor the emerger until the EN the interve t problems. reBrigade, T cut of CRYC nd the UPS er fault for th n the variable ervention. T tions in Mey	with databa d fixed quic cted due to a received. (Ex remotely, or ver electrical 1502/55 swit AS, the trip of SS system to le for "gas d by CMS crew e extraction on the UPS f UPS, since it nfirms a bad ncy stop, bui I-EL piquet of ention went, II, EN-EL piq O. It should allowed for the stop of t yrin.	AUG on 3.3KV tractable Circuit 1-site intervention consumption. chboard: Interlock caused the CMS of the ventilation etection". The r on the DSS for the ventilation for CRYO in the 's very likely the smell around the t to stand by to can come and put , batteries were guet and CRYO let be noted that the early detection. ircuit "FAIR" in ntroller for the che cooling tower			
Plans	EN-CV has a prob It was agreed in th operators available should normally no when opening the	lem with le TIOC to e in case ot be any electrical	a valve in o plan this of a cut a impact by cabinets.	SF2 (LHC2) s for Monday nd it's still be this interve IMPACT nu	) that require v 21/8. CRYC efore restart ntion, but the umber: 21568	es an intervention. ) will have of the LHC. There e risk is present 5			

Intervention Request									
Yes / No	/ No Duration Preferred date/time								
Reason									
Impact									

Linac 4								
Machine Coor	dinator last w	/eek	Alessandra Lombardi					
Machine Coor	dinator this w	veek	Giulia Bellodi					
Statistics								
Availability	99.9%							
			Facility Status					
	A very good week.							
Summary	-On Wednesd see if for diffe after the RFC transmission. can be scann at the source -the elevator wednesday	day so erent s ) impr Prob led as test s door	burce test for 1 hour: Change LEBT gas injection pressure to space charge neutralization degrees the beam pulse flatness roves. It is possible to achieve a flatter pulse at the expense of ably room for improvement if solenoids and steerers' values is in MD mode. Very useful test to be followed up with studies stand. was fixed, profiting from the stop for PSB intervention on					
Issues	-trip of CCDT -BIC stopping converter, no communicatio	Lon S the s t con on. So	Sunday, source on wrong signal from L4L.NFH.014 (old Einzel lens nected) that auto-calibrated periodically and caused missing blved on Wednesday by putting the converter into Blocking.					
Plans	Continue ope	ratior	n as usual.					
			Intervention Request					
Yes / No	Duration		Preferred date/time					
Reason								
Impact								

	PS Booster										
Machine Coor	dina	ator last week	F. Asvesta								
Machine Coor	dina	ator this week	S. Albright								
			Beam S	cheduled							
ISOLDE	Ye	S		PS	Yes						
		Beam	Availability	by Destination (A	AFT)						
ISOL DE	92	%		PS	89.4 %						
	02		Facilit	v Status							
	•	Challenging w	eek requiring	two accesses or	Monday for the regular						
		inspection of C	QFO11 and o	n Wednesday for	the replacement of						
Summarv		BI1.BSW1L1.4	311.BSW1L1.4. Many thanks to everyone involved during the various								
,		Interventions and stops.									
	•	Verification of	the HiRadMa	at beams in the PS	SB.						
	•	Planned acces	s on Monda	y morning for the i	nspection of QFO11. Situation						
		on the quadru	oole remains	stable.							
		In parallel:	ra installed to	monitor the wate	r leak on QEQ11						
		<ul> <li>Inspective</li> </ul>	tion of BI.BS	W: A water leak w	vas found on the coil of						
		BI1.BS	SW1L1.4. Th	e experts allowed	the restart of the machine,						
		nowev coil as	soon as pos	a scheduling a sto	op for the replacement of the						
		during	FOM, consid	dering the concern	ns of the full complex.						
		Full access and recovery took about 2h, 1 for the regular inspections and to assess the BI1 BSW11 1.4 situation									
		to assess the l	dnesday mo	.4 situation.	e leaking coil on						
		BI1.BSW1L1.4	Intervention	n was successful	and the experts identified the						
		leak where they expected. However, when reinstalling the magnet some									
		condensation	was observe	d on ring 3. The re	eason for this is not yet known						
		be accessing t	he machine	during the regular	inspections for the						
		quadrupoles to	o follow up th	e status.							
		In parallel:	onal lighting i	nstalled to facilitat	te the remote checks for the						
Issues		water	leak on QFO	11 as the recordir	ng was too dark. Indeed, the						
		situati	on is much c	eaner now and th	e camera can be used to						
		o Measu	ely observe ti irements on t	ne leak. the machine quad	rupples for other possible						
		water	leaks. The m	agnets team cheo	cked thoroughly magnets						
		showii	ng suspicious	measurements,	and no other leaks were found.						
			ar inspection ard replacen	s will continue as nent on BR2.QCD	planned. .1.						
		∘ RF an	plifier replac	ement for B-train	system in ring 4 (not used in						
		operat	ions).	d the emilities and th	- Diamond DI Majanala						
			am removed a only the ar	notified ones for a	e Diamond BLM signals, ill rings, OP verified the signals						
		with b	eam.								
		Activities took	overall 10h4	5min. Vacuum rec	covered fast in ring 1 and						
		During the res	tart:		<i>ε</i> α.						
		<ul> <li>The di</li> </ul>	stributor, BI.I	DIS10, was in fau	t and could not be reset from						
		the CO	CC. Both the	piquet and the ex	pert were contacted and						
		identif thresh	ieu me sourc olds were ch	anged and beam	was in the PSB after an						
		additio	onal 1h15. Th	e ABT team is fol	lowing up the situation to						

	0	<ul> <li>the access.</li> <li>After about 30min of normal operation, an interlock on the H0H-monitor for R1 occurred. The interlock was present even without beam and the experts were needed onsite. A large offset on the plates signal seems to be the reason for the interlock. The experts exchanged a card, but the problem persisted, while it was resolved once the old card was put back in place. The reason for this offset or how it was resolved is not clear and the experts are closely monitoring the situation. This problem blocked operations for an additional 4h8min.</li> </ul>						
Plans	Normal oper	rations, providing be	eam to downstream ma	achines, experiments and				
	MDs							
	-	Interventi	on Request					
No	Duration		Preferred date/time					
Reason								
Impact								

ISOLDE									
Machine Supe	ervisor last wee	k	Lefteris F	adakis					
Machine Supe	ervisor this wee	k	Simon M	ataguez					
Beam Scheduled									
GPS	Yes	HRS		No	HIE-ISO		No		
	Beam Availability by Destination (AFT)								
GPS	%97	HRS		%	HIE-ISO		%		
Facility Status									
Summary	Faced issues v started.	ith the	e HT at the	e start of the run	but run w	ent smoot	h once they		
Issues	Issues with bot need to investing We were not all the clamps. Ex us to carry on.	h HT p gate al ple to p perts r	bower sup nd decide put in plac made a fas	olies. GPS was on course of act e the target in H st intervention to	solved eas tion. RS due to change th	sily but HF a fault or ne piston a	RS we still the piston of and allowed		
Plans	Find a slot to re CRIS experime	epair th ent.	ne HRS H	T. Work on settir	ng up the	beam fron	n HRS to the		
			Interventi	on Request					
Yes / No	Duration			Preferred d	ate/time				
Reason									
Impact									

PS									
Machine Coor	dinator last	week Al	ex Huschaue	er					
Machine Coor	dinator this	week Al	ex Lasheen						
Beam Scheduled									
East Area	Yes	nTOF	Yes	AD	Yes	SPS	Yes		
		Beam Ava	ailability by	Destinatio	n (AFT)				
AD	75.3 %	EA N	76.1 %	EA T8	76.1 %	EA T9	76.1 %		
nTOF	73 %	SPS	74.2 %						
			Facility S	Status					
Summary	<ul> <li>Low intensity TOF cycle at 20E10 ppb sent to TOF Monday-Tuesday night</li> <li>SFTPRO intensity reduced to ~600E10 ppp on SPS-NA request</li> <li>Sublimation started every day at 14h in view of ion run</li> <li>Limits of EAST proton sharing discussed with physics coordinator(s) and at user meeting following frequent requests to increase the intensity per spill or the number of spills</li> <li>Tests with wire scanner in SS54 performed with expert – can be used during working hours, preference to be given to BWS68 to measure MTE beam</li> <li>SFTPRO high-intensity tests up to 3300E10 ppp</li> <li>Hardware checks in preparation of ion run (cavities pulsing and synchronization ok for EARLY and NOMINAL beams)</li> </ul>								
Issues	<ul> <li>Acc MH.</li> <li>Sev</li> <li>Vac SSC</li> <li>KFA Mor</li> </ul>	ess Mond z ok and r eral interv Internal d CPU boar with settir Additiona uum leak 03: Aluminum bypass ca Alumost 17 VSC and Second R KFA04 ar s of some ace gap r ier-bucke unday, lo v71 modu aday	lay morning t rephased, wo ventions prof ump water flund for PE.SM ng from a diff I work on 20 developed T in seal showe aused by SF <sup>T</sup> Th of downtin RF teams for RF bypass in nd BFA09 co a 10 MHz cav relay of 10 MHz cal reboot by le 11 not ope	o repair 20 ork ongoing iting from P ow errors re H16 replac erent user MHz cavity hursday ev d burnt spo TPRO high- ne for vacu r rapid inter SS02 preve nditioning o rities Saturo Hz cavity in o n SFTPR the piquet erational, ex	MHz and 40 for 20 MHz) SB magnet r esolved ed to address r in SS92 rening at the at, very likely intensity test um recovery rvention!) entively repla of ~1h required day afternoor SS66 CO during the required spert will cont	MHz amp eplaceme s pulses o downstrea due to dar ts (many tha ced as we ed prior to n, access r a night from tinue to wo	Alifiers (40 Int: Int: Courring Am flange of Maged RF Inks to RP, All restart required to In Saturday Drk on		
Plans									
		I	Intervention	Request					
Yes	Duration	1h30	P	referred da	te/time Ear	ly in the w	/eek		
Reason	- Gap	relay of l	PA.C10.66 n	eeds replac	cement	roliobility			
Impact	No beam fo	r users an	nd downstrea	m machine	S	Glability			

PS East Area									
Facility Coordinator last week         N. Charitonidis									
Facility Coord	linator this	week	B. F	Rae					
				Beam Sch	eduled				
<b>T</b> 8	Yes	<b>T9</b>		Yes	<b>T10</b>	Yes	T11	No	
Beam Availability by Destination (AFT) General: 83.3%									
Running T8	83.3%	<b>T9</b>		83.3%	<b>T10</b>	83.3%	T11	N/A	
Facility Status									
Summary	T09: No iss T10: Brief i T11: No us	sues, sn ssue wi er.	nooth th ma	n operation agnet T10.	BHZ017, fixe	ed with reset	. Smooth op	peration.	
Plans	<ul><li>T9: EN</li><li>T10: A</li></ul>	UBET, LICE IT	EP3I S3 ->	DET contin > EIC dRIC	ue. H (Tuesday	22-08-2023	)		
			In	tervention	Request				
No	Duration			P	referred dat	te/time			
Reason									
Impact									

PS nTOF							
Facility Coord	linator last w	veek Michael Bac	ak				
Facility Coord	linator this w	week Michael Bac	ak				
	Beam Requested						
Yes							
Facility Status							
Summary	Progressing	with physics progra	amme according t	o planning.			
Issues	No issues o	on experiment side					
Plans	<ul> <li>EAR1: 243Am(n,f) out; PPAC particle identification test</li> <li>EAR1-NEL: R2E – SEL with high energy neutrons &amp; TOF</li> <li>EAR2: 26Al(n,p) and (n,a)</li> <li>NEAR: preparation of next irrad, campaign</li> </ul>						
		Foreseen	Beam Stop				
Yes	Duration	6h	Date/Time	WED 23.08.23; 10h00			

	AD - ELENA									
Machine Supe	ervisor last week	Br	Bruno DUPUY							
Machine Supe	rvisor this week	Se	ergio PAS	INELLI						
Beam Scheduled										
AD	Yes		ELI	ENA	Yes					
Availability (AFT)										
AD	97.2%		ELI	ENA	97.2%	/ 0				
Facility Status										
Summary	AD and ELENA are in nominal operation mode. However, the yield of the antiprotons is strongly impacted (around 10%) by the proton beam position on the AD TARGET (angle and position). Yan Dutheil installed a stabilizer, and started measures to find the causes of this fluctuation in the ETA and DL lines.									
Issues	<ul> <li>Few trip of the or needed local fi</li> <li>Power supply is work, fixed by the Focalization Hoservice" for this d for a local RESE</li> <li>Wednesday, the future failures.</li> </ul>	DR.QU rst line ssues o first lin orn was evice).	AD powe intervention in the AD e. down by To avoid cavity tu	r supply. Sometir on. electron cooler s safety interlocks wasting more tim be amplifier has	me rec solenoi Friday ne, the been c	overy by remote reset d, outside of regular / at 2 a.m. (no "standby supervisor has come changed to prevent				
Plans	Operation as usu	al								
		Inte	ervention	Request						
Yes / No	Duration			Preferred date	/time					
Reason										
Impact										

	SPS									
Machine C	oordinator last	week	Kevin Li	Kevin Li						
Machine C	oordinator this	week	Carlo Zannini							
			Beam Scheduled							
LHC	No N/	4	Yes	AWAKE	No	HiRadMat	No			
	I	Beam	Availability I	by Destination	on (AFT)					
LHC	- N/	4	70%	AWAKE	-	HiRadMat	-			
			Facilit	y Status						
Summary	Facility Status         A relatively poor week for the SPS, especially given that we should be running in a quiet and stable period – overall availability only a lean 70%. The original plan was NA production physics and three days of LIU beam development and commissioning on Wednesday, Thursday and Friday.         The Wednesday MD had to be cancelled to allow for a 24 h intervention on one of the Booster magnets of Ring 1. This led to a 12 h downtime for NA physics beams. Beam was re-established Thursday morning, and the LIU MDs could then take place as scheduled. The MD was cumbersome, due to initially poor beam quality from the PS (longitudinal splitting and coupled bunch instabilities) but also due to the SPS 800 MHz cavities running very unreliably with many faults and trips. The RF team has been investigating all day long and has added additional diagnostics to help understand the problem, which is currently not yet identified. The MD was stopped due to a vacuum interlock on some kickers when bunch lengths got comparatively short at high intensities close to flat top. When recovering from the LIU MD on Thursday evening it turned out, that the PS had experienced a vacuum leak in an RF bypass, which consequentially had to be exchanged. The intervention plus subsequent pumping took the machine down for another 17 h. Followed by some problems in the SPS with SBDS electronics, beam was back only after noon. LIU studies continued then, and eventually reached back the 2.2e11 pb at 1.6 ns at flat top. The refurbished wirescanners have held up so far withstanding beams reaching up to 2.3e11 pb at bunch lengths of around 1.6 ns. The mitigations methods to suppress excessive wire heating with high intensity beams seem to be working so far.									
	channelling now	v runni	ng routinely in	n conjunction	with the A	BO noise sup	pression.			
Issues	<ul> <li>800 MF fault an</li> <li>SBDS f</li> </ul>	alysis	to be done. ling threshold	d needs to be	e reverted b	ack to 400 de	egrees.			
Plans	<ul> <li>HiRadM</li> <li>Dedicat</li> <li>HW cor</li> </ul>	/lat bea ted MD nmissi	ams next wee ) on Wedneso oning for ions	ek day s from Thurso	day					
		1	Interventi	on Request						
No	Duration			Preferred d	late/time					
Reason										
Impact										

	SPS North Area								
Facility Co	ord	inator la	ast we	eek N	N. Charitonidis				
Facility Co	ord	inator th	his w	eek E	3. Rae				
Beam Scheduled									
H2	Ye	s	H6		Yes	K12	Yes	P42	Yes
H4	Ye	S	H8		Yes	М2	No	TT20	Yes
Beam Availability by Destination (AFT) General: 71.7%									
H2	71.	7%	H6		71.7%	K12	71.7%	P42	71.7%
H4	71.	7%	H8		71.7%	М2	71.7%	<b>TT20</b>	71.7%
					Facil	ity Status			
Summary		H4: Nor H6: Nor H8: Obs good op M2: No P42/K1 request Sharing as NA62 the inter the H6 a	mal c mal c serve beratio user 2: Go of N/ g: Fro 2 runs nsity and H	operatio operatio d again on. ood bea A62. om Wed s at 22 can be l8 users	n. No issues n. Negative l electrons at m operation. nesday on, v units at T10 adjusted to r s.	beam given -288 GeV/c Running at ve would like and MUonE natch the 22	T4 wobbling during wobl lower intens to request is starting ir units on T1	y change. bling change sity of 22 unit 50 (T2) - 37 n the M2 bea 0, which is t	No issues, s on T10 on (T4) - 50 (T6) mline. For T4, ransparent for
Issues		H2: Movineed to P42/K12 triggere	ving b be id <b>2:</b> So d P0s	beam is lentified me few survey t	sue still pres I, SY/EPC fo times magn to close the F	ent due to m llowing. et currents ra 242 TAX (fev	agnet / pow an out of ref v minutes de	ver converter erence for a owntime).	problems that spill, which
Plans	PlansH2: HIKE SAC $\rightarrow$ LHCb ECALH4: FASER NU $\rightarrow$ RD51/GIF++H6: ATLAS ITK PIXEL $\rightarrow$ MONOLITH, EP PIXELH8: AMS L0, PAN, SND/LHCb $\rightarrow$ LHCb, SELDOM (TWOCRYST), SND/LHCbM2: No user $\rightarrow$ MuonE							ND/LHCb	
					Intervent	ion Reques	ts		
Yes		Duratio	on	2hrs	Preferre	d date/time	Monday (TBC)	/ 21 <sup>st</sup> mornii	ng or afternoon

SPS AWAKE						
Facility Coordinator last week		veek	Giovanni Zevi Della Porta			
Facility Coordinator this week		-				
Facility Status						
Summary	Limited access in TAG41 due to GSM cable installation. Small interventions					
Issues						
Plans	Vapor source tests with laser for diagnostics calibration. Prepare for protons starting on 26 August (or 28 August, depending on HiRadMat)					
Foreseen beam stop						
Yes / No	Duration			date/time		

LHC						
Machine Coordinator last week		k J. Wenninge	J. Wenninger & M. Solfaroli			
Machine Coordinator this week		k J. Wenninge	J. Wenninger & M. Solfaroli			
Statistics						
Availability	NA		Stable Beam Ratio	NA		
Facility Status						
Summary	At the beginning of the week, cryo started filling the arc magnets with liquid Helium. By the end of the week, all sectors filled and at 2K, last adjustments in the arc for S78. LSS8L and ITL8 ready.					
Issues						
Plans	Cryo ready for powering. ELQA of ITL8 and main arc circuits. Start of powering tests. Machine closed end of the week.					
Intervention Request						
Yes / No	Duration		Preferred date/time			

Linac 3					
Machine Supe	rvisor last week	G. Bellodi	G. Bellodi		
Machine Supe	rvisor this week	D, Küchler	D, Küchler		
Statistics					
Availability	NA				
Facility Status					
lon species	lead				
Summary	<ul> <li>LBS measurements and debuncher setup on Monday.</li> <li>Final beam characterization with reference beam profile measurements taken on Tuesday</li> <li>Ramping cavity scan and beam energy measurements repeated on Wednesday after RF phases were accidentally lost the day before.</li> <li>Beam delivered to LEIR on Wednesday as scheduled.</li> <li>~35uA delivered on average on ITL.BCT25 since</li> </ul>				
Issues	<ul> <li>Observed a ~kHz oscillation on shot-to-shot ITL.BCT05 without beam.</li> <li>BI following up.</li> <li>Buncher and ramping cavity phase settings lost after RF restart</li> </ul>				
Plans	Normal operation and beam delivery to LEIR				
Intervention Request					
Yes / No	Duration		Preferred date/time		
Reason					
Impact					

CLEAR				
Facility Coordinator last week		Vilde Rieker		
Facility Coordinator this week		Wilfrid Farabolini		
Facility Status				
Summary	<ul> <li>3rd (and last) week of the summer-shutdown</li> <li>Main tasks related to the re-commissioning of the plasma lens, which involved vacuum intervention.</li> <li>Succesful test of the plasma-lens (without beam).</li> <li>Identified locations for future BBP experiment and installed fibers.</li> <li>Attempt to re-start klystrons and magnets for a quick test towards the end of the week failed as is was not possible to get the necessary permissions in time.</li> </ul>			
Issues	RF interlock on klystrons and main switch for magnet power converters consigned.			
Plans	Re-start of CLEAR and run the planned plasma lens experiments.			